Author's response to reviews

Title: Efficacy of Acupuncture and Electroacupuncture in patients with nonspecific low back pain: study protocol for a randomized controlled trial

Authors:

Josielli Comachio J.C (josiellicomachio@usp.br)
Mauricio Oliveira Magalhães M.O.M (m.magalhaes@usp.br)
Thomaz Nogueira Burke T.N.B (thomazburke@gmail.com)
Luiz Vidal Ramos L.A.V.R (luiz.armando@usp.br)
Jecilene R Costa-Frutuoso J.C.F (jecilenecosta@hotmail.com)
Amélia Pasqual Marques A.P.M (pasqual@usp.br)

Version: 5 Date: 21 June 2015

Author's response to reviews: see over

Editor
Trials

Your reference: Manuscript Submission

We are resubmitting our article entitled: “Efficacy of Acupuncture and Electroacupuncture in patients with nonspecific low back pain: protocol of a randomized clinical trial”, TRIALS as original article. We also confirm that our manuscript has not been submitted and is not simultaneously being submitted elsewhere.

This study had Public Financial Support of: (CAPES) - Coordenação de Aperfeiçoamento de Pessoal de Nível Superior – National Consult of Research and Development and there is not conflict of interest to the work.

Regarding the paper, we had submitted to Trials, we sincerely thank the suggestions put forward, as they neatly contributed to improving our study.

Please find enclosed the article final version, changed in accordance to the suggestions made, as well a letter containing our replies.

While waiting for the journal new assessment of our work, we remain.

Yours sincerely,

Author